CENWP-OP-B                                                                                                                          29 January, 2007
MEMORANDUM FOR CENWP-OP, ATTN: Operations Biologist
SUBJECT: Bonneville Lock and Dam, Fishway and Fish Activities for Week 04 of 2007, which covers the period from 21 through 27 January 2007.
1 OPERATION SUMMARY

1.1 Daily average river flows ranged from 158.9 to 193.4 kcfs.  Daily average project tailwater elevation ranged from 14.6’ to 16.1’ msl.  Daily average powerhouse forebay elevation ranged from 72.2’ to 74.5’ msl.  Daily average river temperature ranged from 38(F to 39(F.  Secchi disk measurements averaged 3.0 to 5.0 feet.

1.2 Spill Bays 1 and 18, which supply attraction flow for Cascades Island and B-Branch respectively, are included in the spill pattern.  Daily average spill averaged 0.5 to 0.6 kcfs.  
1.2.1 Spill Bay 1 remains closed during all hours.  

1.2.2 Spill Bay 18 is open during daylight hours.
1.3 Unit Operation.  
1.3.1 Unit 2 remains out of service for turbine rehab.
1.3.2 Unit 8 remains out of service for turbine rehab. 

1.3.3 Unit 10 remains out of service for turbine rehab.
1.3.4 Unit 11 remains out of service for VBS modifications.
1.3.5 Unit 12 was out of service from 1432 until 1611 on 25 January for a packing adjustment.  

1.3.6 Unit 13 remains out of service for a four year overhaul.

1.4 Second Powerhouse fish units provide attraction flow for the Washington Shore fish ladder.  The units are currently out of service for winter maintenance. 

1.5 Gatewell drawdowns are measured on Wednesdays; VBS drawdowns are measured daily on modified units.
2 ACTIVITIES

2.1 AWS Closures.  The following AWS valves are closed.
2.1.1 FV5-3, FV5-4, and FV5-9 remain closed for winter maintenance in the Cascades Island fishway.

2.1.2 FV6-9 remains closed for winter maintenance in the Washington Shore fishway.

2.1.3 FV 3-9 was closed for cleaning from 1502 until 1544 on 22 January.

2.2 STS/VBS Inspections.  STSs/VBSs have been removed for winter maintenance.  Screens will be installed the last week of February.
Table 1.  STS and VBS inspections.

	Unit
	Date
	Run Hours
	Comments

	11-18
	NA
	NA
	All screens removed for winter maintenance.


2.3 Dewatering and Fish Salvage.  No dewatering or fish salvage activities occurred this week.
3 RESEARCH
3.1 ERDC TDG.  Sensors remain installed in the tailrace.

3.2 Fish Field Unit.  Pinniped observations began on 11 January.
3.3 NOAA.  Construction of the lamprey ramp in the WA shore AWS began this week.  
3.4 USGS.  Lamprey held in the FERL for tagging evaluation.
3.5 WDFW Fish Counting.  Video counting continues at Bradford Island.
4 FISHWAYS 

4.1 Project biologists inspected 22 through 25 January 2007.
4.2  Fish Passage Plan Observations.
Table 2.  FPP Out of Criteria items.

	DATE
	LOCATion
	FPP itEMS
	CAUSE
	RESPONSE
	TR#

	BRADFORD ISLAND AND BRANCHES

	1/22
	B-Branch 
	Entrance differential < 1.0
	System in manual
	Operators adjusted.
	

	1/22, 1/23
	Bradford Is.
	BI weir < 1.0
	FV 3-9 not controlling properly.
	TR
	07-15505

	1/22, 1/23
	A-Branch
	Weir < 1.0
	FV 3-9 not controlling properly.
	TR
	07-15505

	1/22
	B-Branch
	Weir < 1.0
	FV 3-9 not controlling properly.
	TR
	07-15505


4.2.1 PH1 electronic velocity meter is in an eddy.  Due to this placement, the velocity meter reads less than 1.5 fps.  Readings to meet FPP criteria are taken from the south end mechanical meter.

4.2.2 PH1 PCC diffusers FG2-22A and FG2-19 are closed or partially open.  Repairs scheduled for 07/08.  

4.2.3 FG3-3 stuck partially open.  Repairs are scheduled for winter 07/08.  TR#07-2660

4.3 Fishway Temperatures.  Temperatures are taken 01 March through 30 November.
4.4 Zebra Mussel Monitoring.  No signs of colonization were observed this reporting week.  

4.5 Adult Fishways. 
4.5.1 Calibration is checked on the weekends.  B-branch tailwater is out of calibration.
4.5.2 B-Branch PLC is in manual.  
4.5.3 WS and CI remain dewatered.
4.5.4 SLEDS are installed at PH1, PH2, and B-branch fishway entrances.  

4.6 Juvenile Fishways.
4.6.1 DSM1 will continue to remain OOS until removal in winter 07/08.
4.6.2 DSM2 remains dewatered.
4.7 Lamprey Fishways.  Remains dewatered.
5 CONSTRUCTION
5.1 B1 ITS Wall Removal.  Plans are to remove the wall, slope the floor, and install three automated gates.
5.2 PH1 STS/VBS removal and TIES removal.  Six TIES and 32 PH1 STSs/VBSs will be scrapped.

5.3 PH2 VBS mods.  Unit 16 is complete.  Unit 11 is OOS.  Unit 15 will be next.
6 HAZMAT, SPILLS AND CLEANUP – nothing to report.
7 TELETYPES AND OTHER NOTEWORTHY ITEMS
7.1 A teletype sent on 22 December details tailwater operations for chum incubation.   Effective as soon as possible, a hard tailwater constraint to operate project tailwater no lower than 13.0 feet during all hours.  One pulse per day with a minimum duration of 20 minutes and minimum elevation of 13.5 feet.  
7.2 A teletype sent on 16 January details low forebay operation on 22 January 2007 to perform aerial surveys of a sandbar near Hood River and shoreline near Wind River.  Previous surveys were cancelled due to weather and water levels.  Request that Bonneville pool be operated as follows:  Hard constraint 71.5 – 72.5 ft msl, soft constraint 71.5 – 72.0 ft. msl.  Effective from 0800 to 1400.   

JAMES R. MAHAR, P.E.

Operations Manager
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